Memo for Exp_Samurai_2017spring

23-Feb-2017 1 Kobayashi T.

works done on 23-Feb-2017

- FDC1
 - •Test of new ASD's (τ =16nsec) using test pulse input
 - oscillating
 - part of bottom ASD's seem to be causing problems, but no improvement yet.
- FDC2
 - gas purge continued
- ICF
 - guide rail fixed on the floor
 - bottom plate for cables fixed on the detector stand
 - gas purge continued
- (KMWPC)
 - vacuum~ 0.7 Pa
 - signal/power/HV cables/gas lines : connected
 - being pumped with bypass